Author’s response to reviews

Title: Psychometric Properties of Instrumented Postural Sway Measures Recorded in Community Settings in Independent Living Older Adults

Authors:

Bader Alqahtani (baa60@pitt.edu;ba.alqahtani@psau.edu.sa)
Patrick Sparto (psparto@pitt.ed)
Susan Whitney (whitney@pitt.ed)
Susan Greenspan (greenspn@pitt.ed)
Subashan Perera (ksp9@pitt.ed)
Jennifer Brach (jbrach@pitt.ed)

Version: 2 Date: 20 Feb 2020

Author’s response to reviews:

Reviewer reports:

A- Eling D. De Bruin, Ph.D. (Reviewer 1):

The Abbreviation mG (milli-gravitational) might be Extended with the Explanation that this relates to the Earth’s gravitational force.

Information has been added, please see P20, L20-21